Author's response to reviews

Title: Protocol for Work place adjusted Intelligent physical exercise reducing Musculoskeletal pain in Shoulder and neck (VIMS): a cluster randomized controlled trial

Authors:

Lars L Andersen (lla@nrcew.dk)
Mette K Zebis (mettezebis@hotmail.com)
Mogens T Pedersen (mtpedersen@ifi.ku.dk)
Kirsten K Roessler (kroessler@health.sdu.dk)
Christoffer H Andersen (cha@nrcew.dk)
Mette M Pedersen (mmp@nrcew.dk)
Helene Feveile (hfe@nrcew.dk)
Ole S Mortensen (osm@nrcew.dk)
Gisela Sjøgaard (gsjogaard@health.sdu.dk)

Version: 4 Date: 15 July 2010

Author's response to reviews:

Dear editor,

We have now changed the format of the tables to fit the journal style, and have made added the sentende about authors approval of the final version.

best regards
Lars